Author's response to reviews

Title: Bias in a protocol for a meta-analysis of 5-HTTLPR, stress, and depression

Authors:

Terrie E Moffitt (tem11@duke.edu)
Avshalom Caspi (ac115@duke.edu)

Version: 2 Date: 18 February 2014

Author's response to reviews:

Dear Editors, As requested, we have:
changed the title
changed the first paragraph
added a title page
added competing interests
added author contributions.
thank you very much, Terrie Moffitt